Author's response to reviews

Title: Whole exome sequencing in an Indian family links Coats plus syndrome and dextrocardia with a homozygous novel CTC1 and a rare HES7 variation.

Authors:

Manjunath Netravathi (sundernetra@yahoo.co.in)
Renu Kumari (reenu.kumari@igib.in)
Saketh Kapoor (kapoor.saketh@gmail.com)
Pushkar Dakle (pushkar.dakle@igib.in)
Manish K Dwivedi (mk.dwivedi@igib.in)
Sumitabho Deb Roy (sumitabhodebroy@gmail.com)
Paritosh Pandey (paritosh2000@gmail.com)
Jitender Saini (jsaini76@gmail.com)
Anil Ramakrishna (anilramakrishna78@yahoo.co.in)
Devaraddi Navallia (devaraddi.navallia@rediffmail.com)
Parthasarathy Satishchandra (drpsatishchandra@yahoo.com)
Pramod Kumar Pal (pal.pramod@rediffmail.com)
Arun Kumar (karun@mrdg.iisc.ernet.in)
Mohammed Faruq (faruq.mohd@igib.in)

Version: 5 Date: 19 December 2014

Author's response to reviews:

Editor’s comments:

"Professional language editing may be helpful since I note that some of the language changes suggested by Dr Jenkinson in her second review have not been done, specifically those mentioned on page 2 and 3 of the second review by Dr. Jenkinson."

We have adopted the editor’s suggestion and incorporated all the language changes as suggested by Dr Jenkinson and have also reviewed our manuscript by a colleague having proficiency in English.